Author’s response to reviews

Title: The Staffordshire Arthritis, Musculoskeletal, and Back Assessment (SAMBA) Study: a prospective observational study of patient outcome following referral to a primary-secondary care musculoskeletal interface service

Authors:

Edward Roddy (e.roddy@cphc.keele.ac.uk)
Irena Zwierska (i.zwierska@cphc.keele.ac.uk)
Peter Dawes (PT.Dawes@uhns.nhs.uk)
Samantha L Hider (s.hider@cphc.keele.ac.uk)
Kelvin P Jordan (k.p.jordan@cphc.keele.ac.uk)
Jon Packham (j.packham@cphc.keele.ac.uk)
Kay Stevenson (Kay.Stevenson@uhns.nhs.uk)
Elaine Hay (e.m.hay@cphc.keele.ac.uk)
Staffordshire Arthritis, Musculoskeletal, and Back Assessment study (SAMBA) team (primary_care_sciences@cphc.keele.ac.uk)

Version: 6 Date: 1 April 2010

Author’s response to reviews:

Dear Dr Norton,

re: The Staffordshire Arthritis, Musculoskeletal, and Back Assessment (SAMBA) Study: a prospective observational study of patient outcome following referral to a primary-secondary care musculoskeletal interface service (Manuscript ID 1142988644350441)

Thank you for sending the manuscript for final proofing. I have made one minor revision - the removal of tracking from table 2. Figure 1 remains unchanged.

Yours sincerely,

Dr Edward Roddy